Lawrence Berkeley National Laboratory

Recent Work

Title

U.S. Energy Service Company (ESCO) Industry: Recent Market Trends:

Permalink

https://escholarship.org/uc/item/3xw2s6ff

Authors

Stuart, Elizabeth Larsen, Peter H. Carvallo, Juan Pablo <u>et al.</u>

Publication Date

2016-09-28

ERNEST ORLANDO LAWRENCE BERKELEY NATIONAL LABORATORY

U.S. Energy Service Company (ESCO) Industry: Recent Market Trends

Elizabeth Stuart,* Peter H. Larsen,* Juan Pablo Carvallo,* Charles A. Goldman,* and Donald Gilligan**

*Lawrence Berkeley National Laboratory **National Association of Energy Service Companies

Energy Analysis & Environmental Impacts Division

Electricity Markets & Policy Group

October 2016

The work described in this report was funded by the U.S. Department of Energy Office of Energy Efficiency and Renewable Energy under Contract No. DE-AC02-05CH11231

Disclaimer

This document was prepared as an account of work sponsored by the United States Government. While this document is believed to contain correct information, neither the United States Government nor any agency thereof, nor The Regents of the University of California, nor any of their employees, makes any warranty, express or implied, or assumes any legal responsibility for the accuracy, completeness, or usefulness of any information, apparatus, product, or process disclosed, or represents that its use would not infringe privately owned rights. Reference herein to any specific commercial product, process, or service by its trade name, trademark, manufacturer, or otherwise, does not necessarily constitute or imply its endorsement, recommendation, or favoring by the United States Government or any agency thereof, or The Regents of the University of California. The views and opinions of authors expressed herein do not necessarily state or reflect those of the United States Government or any agency thereof, or The Regents of the University of California.

Ernest Orlando Lawrence Berkeley National Laboratory is an equal opportunity employer.

U.S. Energy Service Company (ESCO) Industry: Recent Market Trends

Prepared for the Office of Energy Efficiency and Renewable Energy Federal Energy Management Program U.S. Department of Energy

Elizabeth Stuart,* Peter H. Larsen, * Juan Pablo Carvallo, * Charles A. Goldman, * and Donald Gilligan**

*Lawrence Berkeley National Laboratory **National Association of Energy Service Companies

Ernest Orlando Lawrence Berkeley National Laboratory 1 Cyclotron Road, MS 90R4000 Berkeley CA 94720-8136

October 2016

The work described in this report was funded by the U.S. Department of Energy Office of Energy Efficiency and Renewable Energy under Contract No. DE-AC02-05CH11231

Acknowledgements

The work described in this report was funded by the U.S. Department of Energy Office of Energy Efficiency and Renewable Energy (EERE) under Contract No. DE-AC02-05CH11231.

We would like to thank Kathleen Hogan (DOE-EERE) for ongoing support of our research into this industry. We gratefully acknowledge Dr. Timothy Unruh, Kurmit Rockwell, and Schuyler Schell (DOE-FEMP) for providing resources to conduct the analysis and write up the results for this research report. We also thank AnnaMaria Garcia, Jenah Zweig, Alice Dasek, and James Carlisle (DOE-WIP) for supporting our research into the ESCO industry including outreach to ESCO executives for collection of the data for this report.

We thank David Birr (Synchronous Energy Solutions), Patricia Donahue (Donahue and Associates) and Chris Halpin (Celtic Energy) for sharing their expertise on this industry. We thank Phil Coleman (LBNL), Jeanna Paluzzi (Colorado Energy Office), Steve Schiller (Schiller Associates), and Robert Slattery (ORNL) for providing valuable information and insight. In addition, Greg Collins (Energy Systems Group), Steve Morgan (Clean Energy Solutions), Paul Poblocki (Johnson Controls) Rick Rodriquez (Siemens) and Terry Singer (National Association of Energy Service Companies) provided valuable comments on an earlier draft of this report. Finally, we would like to gratefully acknowledge key staff at the ESCOs who spent a considerable amount time responding to our requests for information. Any remaining omissions and errors are the responsibility of the authors.

Table of Contents

Ack	nowled	lgements	1
Tab	ole of Co	ontents	2
Tab	ole of Fi	gures	3
List	of Tab	les	3
Acr	onyms	and Abbreviations	4
Exe	cutive	Summary	5
1.	Introd	uction	11
2.	Data S	ources and Methods	14
	2.1.	Data Sources	14
	2.2.	Method	15
3.	U.S. ES	SCO Industry Revenue and Market Trends	17
	3.1.	Current and historic revenue	17
	3.2.	Short-term revenue projections from ESCOs	
	3.3.	ESCO industry revenues by market segment and size of ESCO	19
	3.4.	Revenue trends over time by type of ESCO business activity and sector	23
	3.5.	ESCO revenues by census region	24
	3.6.	ESCO performance contracting revenues from new and existing customers	26
	3.7.	Incorporation of non-energy benefits in ESPC	27
	3.8.	Financing tools and incentives	31
4.	Discus	sion	35
	4.1.	ESCO industry revenue estimates	35
	4.2.	Factors influencing recent ESCO industry market activity	35
	4.3.	Prospects for near-term growth	41
5.	Conclu	ision	42
Ref	erences	5	44
App	oendix .	A. Data Tables	48

Table of Figures

Figure ES - 1. Reported and projected ESCO industry revenues (nominal \$): 1990-2017	6
Figure ES - 2. Use of non-energy benefits in performance-based projects (2012-2014)	8
Figure ES - 3. Financing Approaches by Market Segment (2012-2014)	10
Figure 1. Aggregate ESCO industry revenue from 1990 to 2014	17
Figure 2. Reported and projected ESCO industry revenues (nominal): 1990-2017	18
Figure 3. ESCO industry revenue share and absolute revenue by market segment	20
Figure 4. ESCO industry revenue share by size of ESCO	21
Figure 5. Share of ESCO industry revenues by market segment and size of ESCO	22
Figure 6. ESCO industry revenue by business activity over time (nominal \$)	23
Figure 7. 2014 revenues by business activity and market segment	24
Figure 8. 2014 ESCO revenues by census sub-region, disaggregated by size of ESCO	26
Figure 9. Contribution to ESCO performance contracting revenues from new and existing customers (2012 2014)	<u>?</u> - 27
Figure 10. Use of non-energy benefits in performance-based projects (2012-2014)	29
Figure 11. Incorporation of non-energy benefits in State/Local Government, K–12 schools, and University/College markets	29
Figure 12. Incorporation of non-energy benefits in Federal government, Healthcare, and C/I markets	30
Figure 13. Use of ESPC strictly for facility improvement purposes	31
Figure 14. Utilization of local, state or federal tax benefits in projects developed by ESCOs during 2012-202	14 32
Figure 15. Project funding sources for performance-based projects by market segment	33
Figure 16. Project financing approaches by market segment	34
Figure 17. Change in tax revenue from example states (Pew Charitable Trust 2015)	38
Figure 18. Debt and unfunded retirement costs as a share of state personal income (Pew Charitable Trust 2015)	39

List of Tables

Table 1. ESCO response rates to requests for information	16
Table 2. Average annual growth rates by size of ESCO	18
Table 3. 2014 ESCO industry revenue by market segment	19
Table 4. Market share of total ESCO revenues received by the eight largest companies since 2006	21
Table 5. 2014 ESCO industry absolute revenue by U.S. Census sub-region and ESCO size	25
Table 6. Number of ESCOs incorporating non-energy benefits in performance-based projects (2012-2014)	28
Table 7. Number of ESCOs reporting incorporation of tax benefits in projects implemented 2012-2014	32
Table A - 1. ESCO industry revenue (nominal \$) by market segment for 2008, 2011 and 2014	48
Table A - 2. ESCO industry revenue share by market segment for 2008, 2011 and 2014	48

Acronyms and Abbreviations

ARRA	American Reinvestment and Recovery Act
Btu	British thermal unit
MMBtu	Million British thermal unit
C&I	commercial and industrial (private sector)
CBECS	Commercial Building Energy Consumption Survey
DOE	U.S. Department of Energy
ECM	energy conservation measure
EERE	(DOE Office of) Energy Efficiency and Renewable Energy
EIA	Energy Information Administration
ESC	Energy Services Coalition
ESCO	energy services company
ESPC	energy savings performance contract
FEMP	U.S. Department of Energy Federal Energy Management Program
HUD	U.S. Department of Housing and Urban Development
HVAC	heating, ventilation and air conditioning
IDIQ	DOE's Indefinite Infinite Quantity (IDIQ) Super ESPC contract
LBNL	Lawrence Berkeley National Laboratory
MUSH	Municipal and state governments, universities and colleges, K–12 schools, and healthcare markets
NAESCO	National Association of Energy Service Companies
NEBs	Non-energy benefits
0&M	operations and maintenance
OE	(DOE Office of) Electricity Delivery and Energy Reliability
OWIP	(DOE Office of) Weatherization and Intergovernmental Programs
PACE	Property Assessed Clean Energy
РРА	Power Purchase Agreement
QECB	qualified energy conservation bond
UESC	utility energy savings contract

Executive Summary

This study presents an analysis of the market size, growth projections and industry trends of the U.S. Energy Service Company (ESCO) industry, drawing on information provided by ESCO executives in late 2015. We define ESCOs as energy service companies for whom performance-based contracting is a core business offering. We identified forty-seven firms that met our definition of an ESCO.¹ Forty-three of these companies responded to our requests for information, representing a 91% response rate.²

We also report 2014 ESCO industry revenues by market segment, region and business activity type, and for new versus existing customers. Finally, we report on use of tax incentives and financing tools, and incorporation of non-energy benefits into performance-based project economics.

We summarize key findings below.

ESCO Industry Revenue and Growth Trends

Key finding: After more than two decades of year-over-year growth, ESCO industry revenues appeared to flatten between 2011 and 2014.

ESCOs reported 2014 industry revenue of approximately \$5.3 billion, which represents no increase over the 2011 ESCO industry revenue of \$5.3 billion (nominal \$) reported by Stuart et al. (2013) (see Figure ES - 1). In an earlier Lawrence Berkeley National Laboratory (LBNL) study (Stuart et al 2013), ESCOs projected annual revenues of ~\$7.5 billion in 2014, which was about 44% (\$2.3 billion) higher than actual ESCO-reported revenues.

Key finding: ESCOs expect total annual industry revenues to be approximately \$7.6 billion in 2017, which equates to an average annual growth rate of ~13% from 2015-2017.

¹For purposes of defining the scope of the ESCO industry, we define ESCOs as firms that provide energy efficiency-related and other value-added services and for which performance contracting makes up a core part of its energy-efficiency services business. In a performance contract, the ESCO guarantees energy and/or dollar savings for the project and ESCO compensation is linked in some fashion to the performance of the project. We exclude companies such as engineering and architectural firms; HVAC, lighting, windows or insulation contractors; companies whose primary business is utility energy efficiency program implementation; and consultants that offer energy efficiency services, but typically do not enter into long-term contracts that link compensation to project energy savings and/or performance. We also exclude companies that only provide on-site generation or renewable energy systems without also deploying energy efficiency measures.

² The four non-respondent ESCOs were small companies in terms of revenue, which we estimate accounted for about 2% of total industry revenues



Figure ES - 1. Reported and projected ESCO industry revenues (nominal \$): 1990-2017

Key finding: Public and institutional market sectors accounted for 85% of industry revenue in 2014, which is consistent with previous results.

The share of revenue generated by the federal sector was $\sim 21\%$ in 2014, the same as in 2011. The share of revenue from state and local projects remained nearly the same between 2011 (24%) and 2014 (25%). However, the percentage of revenue from K–12 schools increased between 2011 and 2014 (from 19% to 23%), while revenue share from the university/college sector declined in the same period (from 14% to 10%). The share of industry revenue from the healthcare sector remained consistent between 2011 and 2014, at 6%.

Key finding: Performance contracting generated 74% (\$3.7 billion) of the industry revenue in 2014, which was somewhat higher than the 69% share from performance contracting in 2011 and 2008.

Design-build projects contributed the next largest portion of revenue in 2014 (16% or about \$800 million), followed distantly by consulting services (5%), onsite generation power purchase agreements (PPA) (3%) and other activities (2%).

Key finding: Large ESCOs' share of total industry revenue decreased somewhat between 2011 and 2014, while medium and small ESCOs' share increased slightly in 2014.

We analyzed share of industry revenue by size of ESCO over time and found that the share of total industry revenues garnered by large ESCOs (annual revenue of \$300M or greater) decreased from 56% in 2011 to 51% in 2014. Medium-sized ESCOs (annual revenues between \$100M and \$299M) increased market share from 29% in 2011 to 33% in 2014. Small ESCOs (annual revenue <\$100M) increased their share of total industry revenue from 15% in 2011 to 16% in 2014.

Key finding: Share of revenue by ESCO size varies for different market segments.

Large ESCOs as a group accounted for 66% of 2014 industry revenues in the federal market, which is higher than their overall market share (51%). However, small ESCOs captured 25% of the K–12 schools market and 29% of the private commercial market, both significantly higher proportions than small ESCOs' overall share of industry revenue (16%).

Key finding: Share of revenue by ESCO size varies in different regions across the United States.

For the first time, ESCOs estimated the distribution of their 2014 revenues from various U.S. Census sub-regions. Disaggregating the results by size of ESCO, we found that large ESCOs accounted for \sim 60-80% of industry revenues in the West North Central, Middle Atlantic and New England regions. However, in the East North Central region, small ESCOs garnered nearly as much of the total market revenue as large ESCOs while in the West and South Central regions, medium and large ESCOs had nearly equal share of revenue (\sim 40% and 45% respectively).

Key finding: New customers accounted for the majority of performance-based revenue during 2012-2014.

ESCOs also estimated the percentage of revenue from performance-based projects generated from new and existing customers in various market sectors. For performance-based projects implemented during the years 2012–2014, new customers³ accounted for approximately 60% of K–12 schools revenue and 85% of public housing revenue. In the federal, university/college and healthcare market sectors, new customers generated about 50% of revenue while 57% of state/local government market revenues were generated by new customers.

³ We defined a new customer as a facility or site the ESCO had not previously provided with energy efficiency or other energy services.

Incorporation of Non-energy Benefits in Performance-based Projects

Key finding: ESCOs reported incorporating at least one type of non-energy benefit in performance-based projects across all markets.

Thirty-eight of thirty-nine respondent ESCOs reported that they incorporated at least one of six types of non-energy benefits (NEBs) in performance-based projects implemented between 2012 and 2014 for at least one market segment.⁴ Nearly all of the ESCOs indicated that three types of non-energy benefits—avoided O&M, avoided capital costs and water conservation— are incorporated across all market sectors [see Figure ES - 2].



Figure ES - 2. Use of non-energy benefits in performance-based projects (2012-2014)

⁴ Incorporation of non-energy benefits (NEBs) into a performance contract involves estimating dollar values for the NEBs and accounting for them in the project economics and guaranteed or stipulated savings.

Tax Benefits and Financing Approaches

Key finding: More than half of the ESCOs that serve each market reported using local, state, or federal tax benefits.

Thirty ESCOs estimated the percentage of their projects implemented during 2012–2014 that used local, state, or federal tax benefits (e.g., Section 179d Investment Tax Credit [ITC], or the Production Tax Credit [PTC]). Across all market sectors, more than 50% of the ESCOs that serve each market reported using tax benefits. Of the ESCOs that responded for the state/local market, nearly 50% (eleven of twenty-four) reported that a high percentage (>66%) of state/local projects used tax benefits. Of the ESCOs serving the K–12 market, 50% (twelve of twenty-four) reported that a high percentage of K–12 projects leveraged tax benefits. In the federal sector, 30% (five of fifteen) ESCOs reported that a high percentage of projects used tax benefits.

Key finding: ESCOs reported use of various financing approaches for projects that were partially or 100% financed, by market.

We also asked ESCOs to estimate the share of performance-based projects (100% financed or a combination of cash and financing) that closed financing during the 2012-2014 period, that used each of the following sources of funds, by market sector: (1) bond; (2) lease; (3) term loan; and (4) other (see right axis of Figure ES - 3). ESCOs reported that most federal projects were financed using term loans. Financed projects in the MUSH⁵ and private commercial markets made extensive use of leases and term loans. Bonds were mostly used for state/local and K–12 schools projects.

⁵ MUSH includes municipal and state governments, universities and colleges, K-12 schools, and healthcare markets.



Figure ES - 3. Financing Approaches by Market Segment (2012-2014)

1. Introduction

The U.S. energy services company (ESCO) industry has a well-established record of delivering substantial, cost-effective energy and economic savings in large and medium-sized facilities, primarily in the public and institutional sector (Vine et al. 1999, Goldman et al., 2005, Hopper et al. 2007, Satchwell et al. 2010, Larsen et al. 2012a, Stuart et al. 2014). The industry achieves significant incremental energy savings each year. In 2012, ESCO-implemented projects in the United States that were still in the contract performance period delivered a total of about 34 TWh of electricity savings. Energy savings from all sources totaled approximately 224 million MMBtu or about 1% of total annual energy consumption in U.S. commercial buildings (Carvallo et al. 2015).

ESCOs deliver most of these energy and cost savings through an energy savings performance contract (ESPC) model. ESPCs are long-term contracts between ESCOs and customers that enable customers to finance energy efficiency, onsite generation and other types of energy projects without the need for significant up-front capital. The ESCO typically guarantees that the project will generate a specified annual level of energy savings sufficient to pay back the project installation and financing costs.

This report builds on previous studies of the U.S. ESCO industry conducted by Lawrence Berkeley National Laboratory (LBNL) and presents policy-relevant findings on recent U.S. ESCO industry growth and market trends. We draw on interviews conducted with ESCO industry executives in late 2015. To define the boundaries of the ESCO industry for purposes of reporting industry trends and market activity, we use the definition of an ESCO established by Larsen et al. (2012a):

A company that provides energy efficiency-related and other value-added services and for which performance contracting is a core part of its energy efficiency services business. In a performance contract, the ESCO guarantees energy and/or dollar savings for the project and ESCO compensation is therefore linked in some fashion to the performance of the contract.

In 2011, ESCOs reported aggregate industry revenues of about \$5.3 billion, with expectations of growing to about \$7 billion by the end of 2014 (Stuart et al. 2014). About 80-85% of industry revenue has come from the "MUSH" market (municipalities, universities, colleges and K-12 schools, state government facilities and healthcare facilities) and federal customers. Historically, ESCO industry growth has been driven largely by enabling policies as well as capital improvement needs of customers. Examples of such policies include the following:

- President Obama called for a combined \$4B in federal ESPC to be implemented from December of 2011 through December 2016 (The White House 2011; 2014). As of May 2016, awarded projects totaled \$3B (Rockwell, 2016).
- Forty-seven states have enabling legislation that allows ESPC (ORNL 2016; NCSL 2013), which allow exemptions from standard bidding requirements, such as the requirement in many jurisdictions that K-12 and local government capital borrowings be approved by

voters, if the ESPC provides the savings guarantee.

- Many of these states offer modest to comprehensive technical support for ESPC implemented in government and educational facilities. For example, the Nevada Governor's Office of Energy administers a comprehensive ESPC program, which includes technical assistance, access to a pool of pre-qualified ESCOs, required documentation and reporting, and financial subsidies for investment grade audits. Several states are in the process of ramping up or re-establishing languishing ESPC programs. However, some previously active ESPC programs in several states have fallen dormant due to budget challenges, complex contracting processes, and staffing reductions in lead agencies that administer ESPC programs.⁶
- Energy efficiency tax benefits and tax deductions, including 179d⁷, enable increased investment and savings for ESPC projects over what would have been possible without the tax benefit. 179d was recently extended through 2016; Stuart et al. (2013) reported that most ESCOs leveraged some form of tax benefits in some of their projects.
- Some states are successfully leveraging Qualified Energy Conservation Bonds (QECBs)⁸ to help finance public sector (city, county, K–12, state) energy efficiency projects, including projects that include performance contracting (EPC 2016). In our interviews, some ESCOs reported that they are informing state energy offices and their local clients about the availability of QECBs.
- Commercial Property-Assessed Clean Energy (PACE) programs⁹ are active in sixteen states (Pace Nation 2016). PACE enables financing of energy efficiency, renewable and water conservation measures in buildings via an assessment on the property's tax bill that can be repaid over a loan term of up to twenty years. Some ESCOs promote the use of commercial PACE to their customers.¹⁰

⁶ California Department of General Services (DGS) reported that the state's current three-contract ESPC requirement makes transaction costs too high for most ESCOs; a 2016 legislative proposal to streamline the ESCO is in process (Sacks 2016). The Arizona Governor's Office of Energy Policy, responsible for reviewing the state's ESPCs, was closed in late 2015 (Randazzo 2015). The Illinois FY16 budget impasse has stalled efficiency projects in K-12 schools and other facilities (Daniels 2016).
⁷ 179d, the Energy Efficient Commercial Buildings Tax Deduction, is a section of the Federal Tax Code enacted under the 2005

Energy Policy Act (EPACT). This section provides a tax deduction to building owners, or to contractors (e.g., architects, design firms, ESCOs) for making energy efficiency improvements to commercial buildings. In 2008, the tax code was amended to allow government agencies to assign their 179d deduction to the implementer or designer of their buildings' efficiency projects. 179d is scheduled to expire December 31, 2016.

⁸ QECBs are U.S. Treasury-subsidized bonds that enable state, tribal and local government issuers to borrow money to fund a range of energy conservation projects at very attractive borrowing rates over long contract terms. QECBs provide subsidies that cover a substantial portion of the interest the public agency issuer pays back to bond purchasers (DOE 2012a). The American Reinvestment and Recovery Act (ARRA) increased the national bond cap for QECBs by \$2.4 billion to a total of \$3.2 billion and provided allocations to each state proportional to population.

⁹ PACE statutes authorize municipalities and counties to work with private sector lenders to provide financing for authorized energy projects (e.g., energy efficiency retrofits, onsite renewable generation) and to collect loan repayment for a term of up to 20 years through an annual assessment on the property's real estate tax bill.

¹⁰ NORESCO (2016) promotes its experience with PACE on its website; Ameresco (2015b) and Johnson Controls (2013b) have publicized projects that leveraged PACE financing.

This report is intended for federal, state, and local policymakers, ESCO industry executives, other energy efficiency service providers and end users. The report provides an explanation of the role that ESCOs can play in securing private capital for energy efficiency investment, including current and projected investment levels, which markets ESCOs reach most effectively and where the potential for growth resides. While the evolution of the U.S. ESCO industry differs significantly from that of other countries' energy efficiency services markets, international policymakers and stakeholders may also benefit from knowledge about U.S. ESCO industry trends to inform their work to expand private-sector energy services industries in their own jurisdictions.

This paper is organized as follows. Section 2 summarizes information about our data sources and analysis approach. Section 3 provides findings on ESCO industry growth and industry characteristics. Section 4 discusses factors that influence recent ESCO industry market activity and prospects for future growth. Section 5 summarizes findings, discusses policy implications, and introduces possible extensions of this work.

2. Data Sources and Methods

In this section, we discuss data sources and methods for developing estimates of 2014 ESCO industry revenues, projected growth in ESCO revenues to 2017, and market characteristics.

For purposes of defining the industry scope and estimating industry revenue, we include only those companies that meet our definition of an ESCO: firms that provide energy efficiency-related and other value-added services for which performance contracting is a *core part*¹¹ of its energy efficiency services business. We exclude firms such as engineering and architectural firms; mechanical contractors that provide a range of energy efficiency equipment installation and energy management services but do not offer performance contracts; HVAC, lighting, windows or insulation contractors; companies whose primary business is utility energy efficiency program implementation; and consultants that offer energy efficiency services, but typically do not enter into long-term contracts that link compensation to project savings and/or economic performance. We also exclude companies that exclusively provide on-site generation or renewable energy systems without also installing measures to address energy efficiency. Some excluded companies serve as subcontractors to ESCOs and may even engage in performance-based work at times, but not as a core business offering. These companies contribute to the broader market for energy efficiency, clean energy and retail energy services, but are not included in our assessment of the U.S. ESCO industry.

2.1. Data Sources

We collected information from several sources including:

- Interviews with ESCO executives conducted in late 2015
- Publicly-available information on corporations' financial performance
- Company websites
- eProject Builder¹²
- The LBNL/ National Association of Energy Service Companies (NAESCO) database of projects
- A Delphi¹³ process with industry experts
- Data from previous LBNL studies that analyzed ESCO industry trends and project

¹¹ We define companies for whom performance contracting is a core part of its energy efficiency services business as companies that self-define as an ESCO in our interviews, those that clearly indicate on their websites that they offer performance contracting, or those that announce performance contracting projects through public news releases ¹² eProject Builder (ePB) is a secure web-based ESPC data management system developed and managed on behalf of U.S. DOE by Lawrence Berkeley National Laboratory. ePB standardizes data collection and reporting for ESPC projects nationwide— across ESCOs and customers in all market sectors. As of September 20, 2016 the ePB database contained 435 projects across federal, state, local, university and K-12 markets. For more information, see eprojectbuilder.lbl.gov.

¹³ A Delphi technique is a process used in business forecasting to reach a consensus via the solicitation and comparison of the views of a small group of experts (Stuart et al. 2014, Satchwell et al. 2010, Hopper et al. 2007, Linstone and Turoff 1975).

performance.14

The primary source of ESCO revenue data used in this report came from communication with U.S. ESCO executives conducted during the summer and fall of 2015. We asked ESCOs to provide information about their company's 2014 revenue from energy services; growth in projected revenue in the next three years; activity by market segment, business type (e.g., performance contracting, design-build, PPA), and region; and the share of new versus existing customers. We defined energy services as performance contracts, energy-efficiency design/build projects, engineering, procurement & construction projects, and energy efficiency-related consulting. For purposes of this report, the definition of energy services excludes retail commodity sales or projects built to supply power to bulk power markets.

We also asked ESCOs to provide information about several topics pertaining to performance-based projects including project financing approaches, use of tax benefits, incorporation of non-energy benefits and typical M&V practices. After reviewing the initial quantitative results, we conducted follow-up interviews with some ESCOs to gain additional insight into some of the key findings.

2.2. Method

We first developed a comprehensive list of firms that might be considered ESCOs for the purposes of this study and used the following sources to identify ESCOs active in the United States:

- NAESCO-accredited ESCOs
- U.S. DOE list of qualified energy service companies for the federal ESPC program
- Members of the Energy Services Coalition (ESC) that indicated that they provided performance contracting
- Qualified ESCOs identified through active state performance contracting programs, including Arkansas, Colorado, Delaware, North Carolina, Oregon and Washington
- Energy services companies identified in previous LBNL market surveys
- Online research to identify other companies who might be engaged in performance contracting; and
- Energy service companies that have requested accounts or training for eProject Builder

Through this process, we identified an initial list of 139 firms that appeared to provide energy efficiency services, possibly including performance contracting. Next, we undertook additional market research on this initial list of firms to determine whether firms were still in business, whether they had been acquired by other ESCOs, and whether they offered energy efficiency

¹⁴ See Stuart et al. (2014), Larsen et al. (2012a), Satchwell et al. (2010), Hopper et al. (2007), and Goldman et al. (2005).

performance contracting as a core activity.¹⁵ Through this process, we identified forty-seven ESCOs that met our criteria and that were still actively working in the United States.¹⁶ Forty-three of the forty-seven companies responded to our requests for information resulting in a 91% response rate, which was somewhat higher than our previous studies (see Table 1).

Year	Respondent ESCOs	Response Rate
Satchwell et al. (2010)	29 of 38	76%
Stuart et al. (2013)	35 of 45	78%
Stuart et al. (2016)	43 of 47	91%

Table 1. ESCO response rates to requests for information

We estimated total 2014 ESCO industry revenue from energy services by summing revenues reported by the forty-three respondent ESCOs and the estimated revenues for the four non-respondent companies.¹⁷ We also developed estimates of 2014 revenues by market segment, region and contract type and for new versus existing customers by multiplying the percent of revenue each respondent ESCO reported for each of these variables by the ESCOs' reported revenue.

We also asked each ESCO to provide annual revenue growth projections for the next three years (2015 to 2017), and developed an aggregate estimate of projected U.S. ESCO industry revenues for that period by applying each ESCO's growth projections to its 2014 revenues.

¹⁵ We excluded fifteen companies included in the previous LBNL ESCO market study because they were acquired by other ESCOs, indicated that they no longer provided performance contracting, or whose websites gave no indication that they provide performance contracting and they did not respond to our requests for confirmation. Of the ninety-two companies excluded from this study, nine had been acquired or were subsidiaries of ESCOs, three appeared to be out of business, twelve responded to us that they are not an ESCO or no longer provide performance contracting, seven were just entering the performance contracting business and could not respond with 2014 or historical data, and 61 appeared to provide some level of energy efficiency services but not performance contracting.

¹⁶ Our market research included: (1) reviewing every company's website to determine whether the company was in business and whether the website included performance contracting among its energy efficiency services; (2) where such information was unclear, contacting the company to ask whether it provides energy efficiency performance contracting; and (3) determining the company was not a subsidiary of another company in our survey. Our initial market research process may not have identified all of the small regional ESCOs or mechanical contractors in the United States that undertook an energy savings performance contract in 2014. Because revenues from energy services for regional companies tend to be modest, we believe that our results represent ~90% or more of U.S. ESCO industry revenue.

¹⁷ For ESCOs that are part of a larger organization, the revenues included for purposes of this report come exclusively from the business unit providing ESCO-related energy services. As part of our QC/QA process, we compared the 2014 results with revenue estimates from the Stuart et al. (2014) study and to other public information (e.g., company reports, U.S. SEC 10-K filings). Estimates of 2014 revenue for the four non-respondent companies came from a Delphi process with industry experts. Total estimated revenue for the four non-respondent companies accounted for only about 2% of aggregate industry revenue.

3. U.S. ESCO Industry Revenue and Market Trends

3.1. Current and historic revenue

We estimate that aggregate U.S. ESCO industry revenue was approximately \$5.3 billion in 2014. For comparison, Stuart et al. (2013) estimated 2011 ESCO industry revenue to be about \$5.3 billion in nominal terms; thus total ESCO industry revenue appears to have been flat (in nominal terms) between 2011 and 2014 (see Figure 1).



Figure 1. Aggregate ESCO industry revenue from 1990 to 2014¹⁸

We calculate average annual growth rates for ESCO firms of varying size as indicated by their annual revenues. For the 2012-2014 period, aggregate revenue for medium-sized ESCOs (\$100M-\$299 per year) grew 4% annually, while revenue for small ESCOs (<\$100M per year) increased by 2.5% annually. The aggregate revenues of large ESCOs (>=\$300M per year) declined by 2.8% annually during this period. In contrast, for the years 2009-2011, revenue for small ESCOs grew 12.5% annually and large ESCO revenue grew at a rate of 10.5% annually (see Table 2). Thus, flat

¹⁸ This figure displays revenue figures only for the years for which we collected ESCO-reported data.

or declining revenues among larger ESCOs appears to be a significant contributor to the overall industry slowdown between 2012 and 2014.

ESCO Size	Average Annual Growth Rate 2009-2011 (nominal)	Average Annual Growth Rate 2012-2014 (nominal)
Small	12.5% (n=32)	2.5% (n=31)
Medium	2.7% (n=8)	4% (n-10)
Large	10.5% (n=5)	-2.8% (n=6)

Table 2. Average annual growth rates by size of ESCO

3.2. Short-term revenue projections from ESCOs

Based on ESCOs' growth projections for 2015-17, we estimate that the U.S. ESCO industry anticipates total annual revenues of approximately \$7.6 billion in 2017, which equates to an average annual growth rate of \sim 13% over the three year time frame (see Figure 2).



Figure 2. Reported and projected ESCO industry revenues (nominal): 1990-2017

We also compare 2014 revenue to industry projections of future revenue from a previous LBNL study (Stuart et al. 2013) in which ESCOs projected that their revenues would be ~\$7.5 billion in 2014 (see orange line in Figure 2). As Figure 2 shows, this revenue projection turned out to be about \$2.3 billion higher (44%) than actual ESCO-reported revenues for 2014 (\$5.3 billion).

The ESCO industry's revenue growth projections for 2015-17 period (\sim 13% annual growth rate) are similar to projections made for 2012-2014 (\sim 12% annual growth rate) [Stuart et al. 2014] but lower than ESCOs' growth projections made for the 2009-2011 period (\sim 25%) [Satchwell et al. 2010].

3.3. ESCO industry revenues by market segment and size of ESCO

We asked ESCOs to estimate the breakdown of their 2014 revenue among various market segments. Forty-three ESCOs provided the information which is summarized in Table 3. In 2014, about 85% of ESCO revenues came from federal, state and local government facilities, universities and colleges, K–12 schools and healthcare and hospital facilities. State/local governments, K–12 schools and federal customers accounted for 25%, 24%, and 21% of ESCO industry revenue, respectively.

Market Segment	Share of Total Revenue	2014 Revenue (\$ million)
State/Local	25.4%	\$1,314
K–12 Schools	23.5%	\$1,219
Federal	20.7%	\$1,073
University/College	10.0%	\$519
Healthcare	5.9%	\$304
Commercial/Industrial	7.9%	\$409
Public Housing/Other	6.6%	\$342
SUBTOTAL (n=43)	100.0%	\$5,180
Non-respondents/Delphi process (n=4)	-	\$95
TOTAL		\$5,275

Table 3. 2014 ESCO industry revenue by market segment

Historically, the bulk of ESCO industry revenue has come from the public and institutional sectors. Figure 3 shows absolute revenue and the breakdown by market sector for 2014 compared to years reported in previous LBNL studies. In 2014, the combined "MUSH" and federal sectors made up 85% of industry revenue, up slightly from 82% in 2011. The university/college sector declined to a 10% share in 2014, down from 16% in 2008. The percentage of revenue generated by the federal sector was 21% in 2014 which is comparable to market share in 2011 and 2006 (22%). The federal market revenue share took a significant, but apparently temporary, dip in 2008, to 15% (Stuart et

al. 2013, Satchwell et al. 2010, Hopper et al. 2007).¹⁹ The private commercial/industrial sector generated about 8% of 2014 industry revenue, the same percentage as in 2011. See Appendix A, Tables A-1 and A-2 for the market share and revenue data presented in Figure 3.



Figure 3. ESCO industry revenue share and absolute revenue by market segment

We also report market share in terms of size of ESCO firm, comparing 2014 results to revenues reported by firms in previous LBNL reports. The share of total ESCO industry revenues going to large ESCOs (annual revenue of greater than \$300M) decreased from 56% in 2011 to 51% in 2014. Small ESCOs have slightly increased market share over this six year time period, from 14% in 2008 to 16% in 2014 (see Figure 4).

¹⁹ Indefinite Delivery Indefinite Quantity (IDIQ) contracts are "blanket" contracts issued to multiple ESCOs by DOE and Army Corps of Engineers. They streamline procurement of ESPC projects by placing them under a single standardized contract (FEMP 2013). The 2007 ESPC IDIQ Continuation Plan required project proposals to kick off by April 1, 2008; projects not meeting that deadline would not be authorized to proceed under the then current DOE ESPC IDIQ contract (DOE 2012b). There was a lull in federal ESPC activity until the contract was extended.



Figure 4. ESCO industry revenue share by size of ESCO

We also observe that the market share of the largest eight ESCOs has declined since 2006. In 2006, these firms accounted for 79% of total revenue whereas the top eight firms accounted for 60% of total industry revenue in 2014 (see Table 4).²⁰

Table 1	Mauliat abaua	aftered ECCO		a a a luca al lace				nnr
Table 4.	iviarket share	of total ESCO	revenues r	eceived by	the eight la	rgest comba	nies since z	LUUb

Year	% of Total Market Revenue	Revenue (\$ million)
2014	60%	\$3,178
2011	70%	\$3,707
2008	76%	\$3,137
2006	79%	\$2,867

²⁰ The eight largest ESCOs (by revenues) changed somewhat year-to-year and includes both large and some medium-sized ESCOs.

We also looked at the activity levels of large, medium and small ESCOs in various market segments and highlight several findings:

- Large ESCOs (>\$300M) accounted for 66% of total revenues in the federal market, which is higher than their overall market share (51%);
- Medium-sized ESCOs' share of revenue in the K–12 (31%), University (35%) and state/local (36%) markets were comparable to their 33% share of total industry revenue;
- Small ESCOs (<\$100M) captured a substantial portion (25%) of the K–12 schools market, which is a significantly higher proportion than their 16% share of total industry revenue; and
- Small ESCOs also do relatively well in the private commercial/industrial market where they account for 29% of the sector's business in 2014 (see Figure 5).



Figure 5. Share of ESCO industry revenues by market segment and size of ESCO

3.4. Revenue trends over time by type of ESCO business activity and sector

ESCOs also estimated the share of their 2014 energy services for various business activities: performance-based contracts, design-build, consulting services, onsite generation power purchase agreements (PPA), utility program implementation and other activities.²¹ Figure 6 shows that the share of revenue from performance contracting in 2014 was 74% (\$3.7B)—this value is somewhat higher than the share (69%) reported in 2008 and 2011. Design-build projects contributed the next largest share of revenue in 2014 (16% or about \$800 million), followed by consulting (5%), onsite generation PPA (3%) and other (2%).²²



Figure 6. ESCO industry revenue by business activity over time (nominal \$)

Forty-two ESCOs estimated the breakdown of their 2014 revenues from each type of business activity by market segment. Performance contracting is the overwhelming choice for government and educational customers when procuring services from an ESCO, but less prevalent in other

²¹ Design- build projects involve fee-based contracts that may include such services as engineering, procurement, project installation and construction; ESCOs do not guarantee energy savings or assume long-term performance risk in these projects. Consulting contracts can include a wide range of activities including audits, engineering studies, project and subcontractor management. Some ESCOs manage or implement programs for utility energy efficiency programs, most commonly in the small commercial or commercial/industrial sector, but occasionally in the residential sector as well. Under a PPA, a third-party (e.g., ESCO) installs and operates an onsite energy generation system and sells the generated energy to the customer.
²² In this study, we asked ESCOs to include utility program implementation services under "Other" rather than as a separate category, and thus that category does not appear in the 2014 pie in Figure 6. It is likely that some revenue from utility consulting contracts, which ESCOs included in the "Utility programs implementation" category in 2011, was reported under the "Consulting Services" category for 2014.

sectors. Performance contracting in the federal sector accounted for \$900 million in 2014, or 85% of the \$1 billion total revenue from these ESCOs' federal customers. Performance contracting generated about \$1.1 billion in 2014 in both the state/local and K–12 schools markets, or 86% and 82% of total ESCO revenue for those markets, respectively. For healthcare facilities, ESCO revenues were split nearly evenly between performance-based and design build projects. Consulting contributed nearly 40% of \$300 million ESCO revenue from public housing and other miscellaneous projects, while design-build dominates ESCO activity in private commercial/industrial facilities, accounting for about 67% of the revenue in that sector (see Figure 7).

PPA/onsite generation contracts were more prevalent in the public housing and commercial/industrial sectors than in the MUSH markets. ESCOs reported no PPA/onsite generation project revenue for 2014 in the federal sector. This was likely the result of a 2012 Office of Management and Budget (OMB) Memorandum that required the federal government to retain title to onsite energy generation equipment installed under an ESPC at the end of the contract term (OMB 2012). ESCOs may be wary that the transfer of ownership to the government will disqualify the ESCO for tax benefits generated by the project.



Figure 7. 2014 revenues by business activity and market segment

3.5. ESCO revenues by census region

For the first time, we asked ESCOs to estimate their distribution of revenues from various U.S. Census sub-regions. All forty-three respondent companies answered this question (see Table 5). ESCOs reported that 20% of industry revenues come from the Pacific region, followed closely by

the South Atlantic (18%) and Middle Atlantic (18%), and East North Central (16%) regions. About 7% of industry revenue comes from New England states, with 6% coming from the West South Central region and about 5% for each of the West North Central, Mountain and East South Central regions.

U.S. Census Sub-region	States	Total Revenue (\$M)	Large ESCOs Revenue (\$M) (% Share for Region)	Medium ESCOs Revenue (\$M) (% Share for Region)	Small ESCOs Revenue (\$M) (% Share for Region)
New England	CT, MA, ME, NH, RI, VT	\$371	\$257 (69%)	\$79 (21%)	\$35 (9%)
Middle Atlantic	CT, PA, NJ, NY	\$899	\$487 (54%)	\$326 (36%)	\$86 (10%)
South Atlantic	DE, DC, GA, FL, MD, NC, SC, VA, WV	\$985	\$542 (55%)	\$388 (39%)	\$55 (6%)
East South Central	AL, KY, MS, TN	\$256	\$109 (42%)	\$79 (31%)	\$68 (26%)
West South Central	AR, LA, OK, TX	\$302	\$157 (52%)	\$130 (43%)	\$15 (5%)
East North Central	IL, IN, MI, OH, WI	\$825	\$375 (45%)	\$178 (22%)	\$272 (33%)
West North Central	IA, KS, MN, MO, ND, NE, SD	\$226	\$181 (80%)	\$30 (14%)	\$15 (6%)
Mountain	AZ, CO, ID, MT, NV, NM, UT, WY	\$286	\$172 (60%)	\$91 (32%)	\$23 (8%)
Pacific	AK, CA, HI, OR, WA	\$1,031	\$410 (40%)	\$422 (41%)	\$199 (19%)
Total – All Regions		\$5,181	\$2,690 (52%)	\$1,724 (33%)	\$768 (15%)

We disaggregate reported revenue by size of ESCO and find that—in the West North Central, Middle Atlantic and New England regions (see Table 5 for states included in each region)—large ESCOs account for ~60-80% of revenue share in each of those regions. However, in the East North Central region, small ESCOs have nearly as much of the total market as large ESCOs. In the West and South Central regions, medium and large ESCOs have nearly equal share (~40% and 45% respectively) [see Figure 8].



Figure 8. 2014 ESCO revenues by census sub-region, disaggregated by size of ESCO

3.6. ESCO performance contracting revenues from new and existing customers

We also asked ESCOs to report estimated share of 2014 performance contracting revenue that came from new and existing customers in various market sectors for the past three years (2012–2014). We defined a new customer as *"a facility or site that your company had not previously provided with energy efficiency or other energy services."* ESCO projects for existing customers typically involve additional project phases or modifications that increase project scope (e.g., retrofit additional buildings at a campus) or install additional measures.

Forty of the forty-three respondent ESCOs provided information on this question. Approximately 60% of ESCO revenues from K–12 schools and 85% of public housing revenue came from new customers between 2012 and 2014. In the federal, university/college and healthcare market sectors, new customers generated about 50% of ESPC revenue while 57% of revenues in the state/local government market were generated by new customers. ESCOs indicated that less than 40% of their revenues in the private commercial/industrial sector came from new customers (see Figure 9).





3.7. Incorporation of non-energy benefits in ESPC

We asked ESCOs to estimate the percentage of performance-based projects implemented during the past three years (2012–2014) in each market segment that incorporated some type of nonenergy benefit (NEB) in the project economics. For this study, we asked about six types of NEBs that are often accepted either in states that authorize inclusion of NEBs, or in federal government ESPC projects: (1) avoided capital costs; (2) avoided operations and maintenance (0&M) costs; (3) benefits from demand response; (4) increased revenues; (5) tradeable emissions credits; and (6) water conservation.²³ 38 of the 39 ESCOs that responded to this question reported including at least one type of NEB in at least one market segment. Table 6 shows the number of ESCOs that indicated that they serve each market sector compared to the number of ESCOs that report quantifying non-energy benefits in that sector.

²³ Avoided capital costs (1) are planned future capital expenditures made unnecessary by the efficiency upgrades. (2) Avoided O&M costs (2) typically include decreased maintenance costs and staffing reductions related to installation of new energy conservation measures. Implementing demand response (DR) measures (3) may not result in overall energy use reductions but DR can reduce a facility's monthly electric demand charges. Increased revenues (4) can accrue if the project includes measures that increase the facility's cash flow (e.g., replacing/upgrading broken parking meters; installing sub-metering that allows the agency to charge utilities to its tenants). In some states, efficiency measures can qualify for air pollution reduction credits (5) monetized through a market-based auction. Water conservation measures (6) can reduce water utility, sewage and other costs.

Market Sector	# of Respondent ESCOs that Serve the Market Sector	# of ESCOs that Report quantifying NEBs in the Market Sector (% of Respondents)
Federal government	17	13 (76%)
State/Local government	31	31 (100%)
K–12 schools	29	28 (97%)
Univ./College	21	21 (100%)
Healthcare	18	18 (100%)
Commercial/Industrial	12	11 (92%)



Nearly all ESCOs indicated that three types of non-energy benefits—avoided O&M, avoided capital costs and water conservation— are incorporated across all market sectors (see blue bars in Figure 10). Demand response benefits were also incorporated across all markets, but to a much lesser degree. A modest number of ESCOs reported that state/local and K–12 projects implemented measures that resulted in increased revenue (e.g., sub-metering that enables agencies to charge utilities to building tenants). A few ESCOs reported incorporating tradeable emissions credits in K-12 projects; emissions credits were rarely or not used in other market sectors (see Figure 10).



Use NEB? No Yes

Figure 10. Use of non-energy benefits in performance-based projects (2012-2014)

Figure 11 and Figure 12 show the number of ESCOs that reported low (1%-33% of projects), moderate (34-66% of projects) and high (>66% of projects) incorporation of NEBs in projects implemented from 2012-2014 for each market segment. ESCOs reported that operations and maintenance (0&M) savings was the most prevalent NEB, captured in at least some projects across all market segments. All ESCOs that serve the K-12 market that responded to this question (27) reported that between 15% and 100% of their K–12 projects incorporated 0&M savings. Avoided capital costs were not uncommon in the MUSH markets, but were not incorporated at all in the federal or private commercial sectors. The Federal Energy Management Program (FEMP) currently permits three types of NEBs in federal ESPC projects: (1) savings due to decreased water and sewer usage; (2) reduced 0&M expenditures; and (3) savings due to reduced repair and replacement expenditures (e.g., less frequent replacement of lighting)[FEMP 2015]. Very few projects across sectors incorporated tradeable emissions credits; a few ESCOs reported emissions credits in state/local, K–12 and commercial projects.



Figure 11. Incorporation of non-energy benefits in State/Local Government, K–12 schools, and University/College markets



Figure 12. Incorporation of non-energy benefits in Federal government, Healthcare, and C/I markets

We also asked ESCOs to estimate the number of performance-based projects that were implemented primarily for facility improvement purposes versus those based on purely financial reasons (energy and other cost savings). Fifty percent of the 28 respondent ESCOs reported that a high percentage of K–12 projects were implemented primarily for facility improvement purposes. Several of these ESCOs reported that 80%-100% of their K–12 projects were completed to address facility upgrade needs. In the state/local government, university/college and healthcare sectors, about one-third of ESCOs reported that a high percentage of those projects were implemented for facility improvement reasons (see Figure 13).



Figure 13. Use of ESPC strictly for facility improvement purposes

3.8. Financing tools and incentives

We asked ESCOs to estimate the percentage of their projects implemented during 2012–2014 that used local, state, or federal tax benefits (e.g., Section 179d,²⁴ Investment Tax Credit (ITC), or the Production Tax Credit (PTC)). Thirty ESCOs responded to this question. Across all market sectors, more than half of the ESCOs serving each market reported use of tax benefits. More than 85% of the respondent ESCOs in the state/local, K–12 and university/college sectors report using tax benefits (see Table 7).

²⁴ 179d allows a public agency to assign the tax credit deduction to the project implementer.

Market Sector	# of Respondent ESCOs that Serve the Market Sector	# of ESCOs Reporting Use of Tax Benefits	
Federal	15	11	
State/Local	22	19	
К-12	24	22	
Univ./College	16	14	
Healthcare	17	12	
Commercial/Industrial	16	11	
Public Housing/Other	8	5	

Table 7. Number of ESCOs reporting incorporation of tax benefits in projects implemented 2012-2014

Figure 14 shows the number of ESCOs that reported zero percent (0%), low (1%-33% of projects), moderate (34-66% of projects) and high (>66% of projects) use of tax benefits in projects implemented from 2012–2014 for each market segment. Of the 24 ESCOs that responded for the state/local market, nearly half (11) reported that a high percentage of state/local projects used tax benefits (see green bar in Figure 14). For the K–12 market, half (12) of the ESCOs reported that a high percentage of K–12 projects leveraged tax benefits. In the federal sector, five of fifteen ESCOs reported that a high percentage of projects used tax benefits.





We asked ESCOs to estimate the percentage of their performance-based projects that closed financing during the 2012-2014 period in each market segment that used the following types of funding approaches: 100% financed; a balance of cash and financing; and 100% cash. Forty-one ESCOs provided information for this question. Figure 15 shows the number of ESCOs that reported zero (0%), low (1-33% of projects), moderate (34-66% of projects), and high (>66% of projects) usage of each funding approach in various market segments. 100% cash funding is most prevalent in the commercial/industrial sector with 44% of ESCOs serving that sector (eleven of twenty-five companies) reporting high usage of cash. Projects in the federal, state/local, K–12, university/college and healthcare sectors had a high prevalence of financing 100% of project costs; more than 50% of ESCOs serving each of those markets indicated high usage of 100% financing (see Figure 15).



Figure 15. Project funding sources for performance-based projects by market segment

We then asked ESCOs to estimate the share of financed performance-based projects (partial or 100% financing) that used each of the following sources of funds, by market sector: (1) bond financing; (2) lease; (3) term loan; and (4) other. Thirty-six ESCOs responded. In the state/local market, term loans are the most commonly used approach with 59% of ESCOs (thirteen of twenty-two) reporting that a high percentage (>66%) of state/local projects used term loans. Customers in the state and local market sectors also make regular use of leases and bonds. Nine of twenty-two ESCOs reported a high percentage of state/local projects using leases, and 35% of ESCOs reported a high percentage of state/local projects using leases, and 35% of ESCOs reported a high percentage in the commercial/industrial sector largely relied on leases or term loans (see Figure 16).



Figure 16. Project financing approaches by market segment

4. Discussion

4.1. ESCO industry revenue estimates

ESCO industry revenues appeared to be flat between 2011 and 2014, and revenue growth projected by ESCOs in 2011 did not occur. Given this result, we undertook additional analysis to review 2014 revenues reported by individual ESCOs and conducted additional interviews with ESCOs and other industry experts to assess factors that may have contributed to this phenomenon. For example, we learned that a few ESCOs that are subsidiaries of large vertically integrated firms had reorganized business units since 2011. Importantly, they indicated that some revenue categories that were included in their responses in our 2011 study²⁵ were not included in their reporting of 2014 revenues. In aggregate, revenues included in the 2011 total, but not in the 2014 figure due to the reorganizations totaled more than \$300 million. Had these reporting changes not occurred, ESCO industry revenues for 2014 would have been somewhat higher than 2011 revenues (in nominal dollars).

The ESCO industry grew steadily from 1990 to 2000, then growth nearly flattened between 2000 and 2004, and resumed again from 2005 to 2011. Several factors contributed to the industry slowdown from 2000 to 2004, including stalled utility retail competition under electric industry restructuring, which led many utilities to divest their unregulated ESCO businesses; fallout from the Enron bankruptcy raising concerns about energy project accounting methods; and, a sunset of enabling federal ESPC legislation in 2003 (Hopper et al. 2007). Similarly we found evidence that a confluence of factors may have contributed to the industry growth slowdown between 2011 and 2014.

4.2. Factors influencing recent ESCO industry market activity

Factor #1: Increased competition from companies that do not meet our definition of an ESCO

LBNL presented preliminary study results to the NAESCO Board of Directors (composed of senior officers of all NAESCO member ESCOs). Based on feedback from some ESCOs, we learned that the industry revenue results likely excluded some revenue from performance-based projects implemented by companies that do not meet LBNL's definition of an ESCO, typically mechanical contractors. We then reached out to ESCO executives on the NAESCO Board of directors and asked two questions:

- (1) To what extent are non-ESCO companies winning competitive solicitations for performance contracting projects?
- (2) What are the names of these non-ESCO companies that they have competed against and that have won projects?

²⁵ Revenue categories included in 2011 that were not included in 2014 results for these ESCOs included demand response, facility management, and renewable energy installations.

Seven ESCOs responded and each company indicated that their companies had lost competitive bids for performance contracts to firms that they would not consider to be ESCOs. The respondents estimated that non-ESCO companies are taking 10-15% of the market and provided a list of twenty-eight mechanical contracting firms that had won projects. Six of those firms met the LBNL definition of an ESCO and were included in the 2014 ESCO industry revenue results. We reviewed the websites of the other twenty-two companies and found that six listed performance contracting as a service (one of the six operates in Canada). The other sixteen companies' websites listed many of the functional components of a performance contract among their service offerings, but did not explicitly name performance contracting as a service.

We then interviewed senior managers from two of the identified mechanical contracting firms to learn more about the mechanical contracting companies that are winning bids for performancebased projects. Both of these managers had more than ten years of performance contracting experience as former employees of large ESCOs and they identified recent changes in the marketplace:

- Some mechanical contractors have become confident that they can manage the risks involved in 12–18 month ESPC project development cycles, which have historically been significant barriers to entry. Some mechanical contractors are increasingly interested in managing their own development risk rather than being subcontractors on projects to ESCOs (and subject to development risks that they believe that they cannot control).
- Some mechanical contractors are now willing to assume the performance risks associated with project measurement and verification (M&V) practices that are typically utilized for certain measures (e.g., stipulated savings or IPMVP Option A).

While this information is anecdotal, we think it suggests that there has been increased competition for energy efficiency procurements in institutional markets in recent years. This trend may also indicate that the ESCO industry has demonstrated that the most common energy efficiency retrofits in the public/institutional market (lighting, HVAC equipment and controls) are no longer perceived as technically risky by many contractors and their customers. Furthermore, an increasing number of customers appear to perceive little risk in the ESCO financing model of relying on the cash flow from projected energy savings to repay the cost of these retrofits. This gradual erosion of perceived technical risk of many high efficiency technologies can be viewed as one of the "success stories" of performance contracting, which has been enabled by federal/state policies and legislation and facilitated by federal (DOE, DOD) and state energy agencies.

A related factor may be the declining cost of emerging technologies (e.g., LEDs, PV) that are being promoted by specialty contractors.²⁶

Factor #2: ESCOs may have already achieved substantial market saturation in key markets

²⁶ We see this as the continuation of a trend we observed in the PV market and declining ESCO revenue from PV installations (Satchwell et al. 2010, Stuart et al. 2013).

The two largest non-federal sources of ESCO revenue have consistently been the state/local government and K–12 schools markets. Though there is significant potential in the private commercial and industrial markets, ESCOs have not had much success in penetrating that market. In our previous study, ESCOs estimated ESPC market saturation of 30% in the state/local market and 42% in the K–12 schools (Stuart et al 2013). These results suggest that for those two sectors, ESCOs may have moved beyond the early adopter phase. It is possible that ESCOs have implemented projects for a substantial portion of their most willing and able customers—akin to "achievable potential."²⁷ The remaining state/local government and K–12 schools market may present higher barriers than the addressed market, for reasons that include the following:

- Smaller projects (<\$500,000), which we hypothesize may make up a large portion of the remaining market and whose owners may be smaller local governments and school districts that historically have not been attractive to the larger ESCOs that dominate the industry.
- A portion of the remaining market is comprised of some potential projects with large technical potential in the K–12 and state/local government markets, as yet largely untapped due to complex bureaucratic hurdles (e.g., New York and Philadelphia K–12 schools, Michigan prisons, and California state government facilities).
- Some of the remaining market may consist of projects for organizations with less financial management capacity than those that have completed ESPC projects.²⁸

Factor #3:Some remaining customers in the public/institutional market are uncertain
about the long-term commitment of an ESPC project

Uncertainty about converting the variable expense of energy bills into the long-term fixed expense of an ESPC financing repayment has always been a significant barrier to ESPC for some customers. The current economic and fiscal environment for state/local, K–12 and public universities, with its budget constraints, uncertain future state revenues, and seemingly unpredictable future energy prices reinforces, rather than reduces, this barrier.

While the U.S. economy has improved significantly since the great recession, the recovery has not been even. Tax revenues have returned to pre-recession levels in only about half of the states. Figure 17 shows the stark difference between one state that has achieved a substantial fiscal recovery (California) and two states that have not recovered (Michigan and Ohio) (Pew Charitable Trust 2015). The experience of the recession has led some states to enlarge "rainy day" funds (to

²⁷ Stuart et al. (2013) defined the ESCO market investment potential as "the aggregate amount of project installation costs technically possible with a single turnover of the remaining stock of buildings not already addressed by ESCOs." The approach to calculating the market potential did not directly align with any of the most common types of energy efficiency estimates (e.g., technical, economic and achievable) due to limitations of the available data. It used a hybrid approach that may have under-estimated the amount of floor space technically addressable, but also potentially over-estimated the amount of economic or achievable market potential.

²⁸ Such organizations include those with financial managers and organization attorneys that are not familiar with ESPC and other types of sophisticated financing.



protect core functions in the event of a future recession).

Figure 17. Change in tax revenue from example states (Pew Charitable Trust 2015)

Furthermore, while state and local government debt has stabilized, the growing burden of retirement and retiree healthcare for public sector employees, which are effectively another form of debt, makes some state and local government increasingly wary of taking on additional long-term debt, such as an ESPC project (see Figure 18).



Figure 18. Debt and unfunded retirement costs as a share of state personal income (Pew Charitable Trust 2015)

Factor #4: Changing perception of future gas and electricity prices

Because government facility managers seek energy price certainty for their budgets, they have used performance contracts to, in effect, hedge a portion of their future costs, by converting a variable expense for energy bills into a long-term fixed expense (e.g., the repayment of the cost of the measures that eliminate a fraction of their energy bills). Expectations that natural gas prices will remain low, and will be significantly less volatile than they were a few years ago, may reduce this perceived benefit of performance contracts for some/many customers.

Larsen et al. (2012b) and Carvallo et al. (2015) report that ESCOs have been installing increasingly comprehensive projects that have a rising share of project savings coming from fossil fuels (as opposed to electricity)—primarily natural gas. The Energy Information Administration (EIA) reports that the U.S. average natural gas retail price for commercial customers rose dramatically from about \$6.50/Mcf in 2003 to over \$12/Mcf in 2006, nearly doubling in three years. Retail gas prices then dropped 30% between 2008 and 2012 to about \$8/Mcf, and have remained near that price and relatively less volatile since then (EIA 2016a). Note that these are the prices for gas delivered to the customer, not the widely-quoted wellhead or trading hub prices. Similarly, the average commercial retail price of electricity 2009-2012 was 10.19 cents/kWh 2009-2011; in 2012-2014 it was only about 2% higher, at 10.36 cents/kWh (EIA 2016b).

Factor #5: Experience with some legacy projects has led to the more cautious use of ESPC to finance needed capital improvements.

Larsen et al (2012b) report that customers in the public/institutional market are increasingly using ESPC projects to help offset accumulated deferred maintenance needs (e.g., asbestos removal, roof replacement, wiring). Such measures are highly valued by public sector customers, but may garner little or no energy savings. If the non-energy benefits cannot be monetized, there may be an increasing gap between projected and actual cash savings during the performance term of the project. Several years ago, when trends indicated that energy prices could inexorably increase, performance contracting customers may have been more comfortable agreeing to stipulating significant energy cost escalations (e.g., 2-3% annually over the life of the contract), because these projected cost escalations translated into more savings and thus a larger project. Escalating the savings in this manner has the effect of showing increased dollar value of savings for projects.

Such cost escalations may be perceived as a barrier for some potential new customers, given that some organizations are struggling to justify the escalations in legacy contracts and to pay the financing costs in the absence of the projected energy cost savings. Minimizing escalations limits the dollar value of the energy savings over the life of new projects and thus the total project cost that can be repaid from savings. Moreover, expectations for continued low natural gas prices have made it more difficult to implement the full scope of projects within payback or maximum contract term constraints. Long-payback measures which may be the major driver from projects (*e.g.*, the new high school roof) are cut from the scope, making the projects smaller and reducing the motivation of the customer to implement a performance-based contract.²⁹

Factor #6: ARRA "boom and bust"

The American Reinvestment and Recovery Act (ARRA) generated an unprecedented short-term burst of funding for energy efficiency investment in the public/institutional sector. Federal, state and local governments were directed to deploy the funds quickly and found opportunities to generate long term energy bill savings by targeting a backlog of "shovel-ready" projects in government facilities (Goldman et al. 2011). Some ESCO respondents indicated that because of time pressures, ARRA funds were often used to pay for short-payback building retrofit projects in public/institutional markets, rather than using ESPC. ARRA also provided funding for state and local governments to hire energy planning and energy efficiency implementation staff and consultants for a limited time. When the ARRA funding ended, many states no longer had funds to retain this staff or consultants who were helping to manage and oversee procurement projects (NASEO 2015).³⁰

²⁹ One reaction to this experience with some legacy projects has been in North Carolina, where in 2012 the state redefined energy savings to severely restrict energy cost escalations (see State of North Carolina Performance Contracting Law, § 143 64.17.(2)).

³⁰ Results of the NASEO 2014 member survey indicated that the median full-time equivalent (FTE) staff at state energy offices declined from 15 in 2012 to 12 in 2014.

4.3. Prospects for near-term growth

There are a number of indications that the ESCO industry may see some growth in revenues over the next few years. A number of enabling policies could have a significant impact on future industry growth in the coming years, including:

- President Obama added an additional \$2 billion to the existing \$2 billion target for federal ESPC investment, bringing the total goal for federal investment in ESPC projects to \$4 billion to be achieved by the end of 2016 (White House 2014). As of May 2016, awarded federal projects total \$3 billion; agency commitments to award by the end of 2016 may add another ~\$1 billion. (Rockwell 2016).
- Several states are ramping up ESPC activity. For example, staff from the California Department of General Services have indicated that the 2016 proposed legislation to streamline the state's ESPC contracting process is expected to pass, which may help address a large backlog of projects that may implemented within the next several years.

It is also important to highlight examples of recent revenue trends for large ESCOs:

- Ameresco, the only publicly-traded U.S. ESCO that is not a subsidiary and whose business consists solely of ESCO services, is experiencing a revenue rebound after several years of decline. The company's reported revenues were \$428M in 2009 and peaked at \$728M in 2011, followed by a 13% decline in 2012 (\$631M) and a further 9% decline in 2013 (\$574M). However, in 2014, revenues rose 3% (to \$593M) and another 8% in 2015 (to \$641M). The company's 2015 SEC 10-K filings indicate that the backlog of contracted and awarded projects (expected future revenues) has increased steadily since 2013 (Ameresco 2010, 2011, 2012, 2013, 2014, 2015; The Street 2011).
- Johnson Controls, Inc. is also seeing revenues rise for its Building Efficiency North American Systems and Service segment³¹ after several years of declines. The segment reported revenue of \$4.6B in 2011 (up 9% from 2010), followed by annual declines of 2%, 1% and 3% in 2012, 2013 and 2014, respectively. In 2015, segment revenue increased by 2% over 2014 (Johnson Controls 2010, 2011, 2012, 2013, 2014, 2015).

³¹ ESCO business revenues for Johnson Controls, Inc. (JCI) account for only a portion of the total revenue for the Building Efficiency North American Systems and Service segment. JCI does not report its ESCO revenue separately from other energy services in its SEC 10-K filings.

5. Conclusion

This study builds on previous LBNL reports on ESCO industry market trends and provides updated estimates of ESCO industry revenue, including revenue by market sector, business type, U.S. Census region, and new and existing customers, as well as recent and short-term projected growth. We also report on the use of non-energy benefits, tax benefits, and financing approaches for performance-based projects.

The U.S. ESCO industry experienced year-over-year growth from 1990 through 2011 with \$5.3 billion in revenues in 2011. However, in 2014, industry revenue remained flat at \sim \$5.3 billion. Based on our interviews, ESCO industry executives anticipate revenues of \sim \$7.6 billion in 2017, which represents an average annual growth rate of \$13% during the years 2015-2017.

The public and institutional market segments have continued to account for the bulk of ESCO industry revenue; in 2014, these markets accounted for about 85% of industry revenues. Performance contracting continues to be the dominant business activity, bringing in ~74% (\$3.7B) of 2014 revenue.

We observed that the share of market revenue going to the top eight ESCOs has declined since 2006. In 2014, in a few U.S. Census regions, small and medium ESCOs captured nearly as much of the market as large ESCOs.

New customers generated approximately 60% of ESCO revenues from K–12 schools and 85% of public housing revenue during 2012-2014. In the federal sector, and each of the remaining MUSH markets, new customers generated 50% or more of ESPC revenue for the three-year period.

ESCOs incorporate several different types of non-energy benefits (NEBs) into performance-based projects. The most commonly incorporated NEBs are O&M savings—which are incorporated in projects across all market segments. ESCOs reported that a significant number of projects, especially in the K–12 schools market, were developed primarily to improve or repair facilities rather than to achieve energy savings.

The ESCO revenue numbers for 2011 and 2014 were not easily comparable, because a few ESCOs that were subsidiaries of larger companies had undergone reorganization; some categories of revenue that were reported in 2011 were not reported in 2014. Other factors that may have contributed to flat revenues between 2011 and 2014 include:

- (1) Increased competition from companies that do not meet our definition of an ESCO (e.g., mechanical contractors), which suggests that customers and contractors are less intimidated by energy efficiency technologies and the notion of financing projects with long-term borrowing that is repaid from savings has spread beyond ESCOs;
- (2) Increased difficulty of developing projects, because in some markets, ESCOs may have attained a substantial portion of the achievable potential, and much of the remaining

market may consist of projects for organizations with less financial management expertise or capacity than organizations that have implemented ESPCs, and large projects for state and local governments that have historically resisted performance contracting, or smaller projects not attractive to the most of the ESCOs in our survey;

- (3) Uncertainty among many potential remaining customers about committing to a long-term performance contract;
- (4) More cautious use of ESPC to finance needed capital improvements by some customers; and
- (5) Reduction of performance contracting's perceived benefit of hedging energy price increases and volatility, because of customer expectations that natural gas (and electricity) prices will continue to be relatively low and less volatile than in the past.

Despite these challenges, recent activity at larger ESCOs and support or expansion of enabling policies (ESPC, PACE) provides some indication that industry revenues are poised to increase in the coming years.

References

- Ameresco. 2010, 2011, 2012, 2013, 2014, 2015. SEC 10-K filings to the U.S. Securities and Exchange Commission. Accessed at: <u>https://www.sec.gov/edgar/searchedgar/companysearch.html</u>.
- Ameresco 2015b. Saginaw Plaza Partners with Ameresco for ESPC to Enhance Energy Efficiency and Upgrade Facility with Renewable Power. Press release August 19, 2015. Accessed at: http://www.ameresco.com/sites/default/files/pace-_saginaw_plaza_-_ameresco_final_8_19_2015.pdf.
- Carvallo, J. P., P. H. Larsen, C. A. Goldman. 2015. Estimating customer electricity and energy savings from projects installed by the U.S. ESCO industry. *Energy Efficiency* 8(6). doi.10.1007/s12053-015-9405-8.
- Daniels, S. 2016. These contractors are out of work because of Springfield standoff. Crain's Chicago Business. March 26. Accessed at: https://www.chicagobusiness.com/article/20160326/ISSUE01/303269992/illinoisenergy-efficiency-program-loses-funding-resulting-in-layoffs.
- DOE (U.S. Department of Energy). 2012a. Making it Easier to Complete Clean Energy Projects with Qualified Energy conservation Bonds (QECBs). Presentation of the DOE Technical Assistance Program. July 19. Accessed at: https://emp.lbl.gov/sites/all/files/making-iteasier-to-complete-clean-energy-projects-with-qecbs-webinar.pdf.
- DOE. 2012b. Department of Energy, Indefinite Delivery Indefinite Quantity, Multiple Award, Energy Savings Performance Contract. Accessed at: http://www.energy.gov/sites/prod/files/2013/10/f3/generic_idiq_espc_contract.pdf.
- EIA. 2016a. Natural Gas Prices. Accessed at: https://www.eia.gov/dnav/ng/ng_pri_sum_dcu_nus_a.htm
- EIA. 2016b. Electricity Data Browser. Accessed at: http://www.eia.gov/electricity/data/browser/#/topic/7?agg=0,1&geo=g&endsec=vg&lin echart=~ELEC.PRICE.US-COM.A~~ELEC.PRICE.US-IND.A&columnchart=ELEC.PRICE.US-ALL.A~ELEC.PRICE.US-RES.A~ELEC.PRICE.US-COM.A~ELEC.PRICE.US-IND.A&map=ELEC.PRICE.US-ALL.A&freq=A&ctype=linechart<ype=pin&rtype=s&pin=&rse=0&maptype=0
- EPC (Energy Programs Consortium). 2016. Table 1A: Qualified Energy Conservation Bonds Known Issued by State (as of February 29, 2016). Accessed at: http://www.energyprograms.org/2016/02/qecb-papers/
- FEMP (Federal Energy Management Program). 2015. M&V Guidelines: Measurement and Verification for Performance-Based Contracts Version 4.0. November. Accessed at: <u>http://energy.gov/sites/prod/files/2016/01/f28/mv_guide_4_0.pdf</u>

- Goldman, C. A., N. Hopper, J. Osborn, T. Singer. 2005. Review of U.S. ESCO market industry trends: an empirical analysis of project data. *Energy Policy*, 33 (2005), 44.
- Hopper, N., C. Goldman, D. Gilligan, T. Singer, D. Birr. 2007. "A Survey of the U.S. ESCO Industry: Market Growth and Development from 2000 to 2006. Lawrence Berkeley National Laboratory 62679. May.
- Johnson Controls. 2010, 2011, 2012, 2013, 2014, 2015. SEC 10-K filings to the U.S. Securities and Exchange Commission. Accessed at: https://www.sec.gov/edgar/searchedgar/companysearch.html.
- Johnson Controls. 2013b. "Johnson Controls and Ygrene Energy Fund sign nation's largest PACE Energy Upgrade Project [Press release]." Accessed at: http://www.johnsoncontrols.com/media-center/news/pressreleases/2013/08/01/johnson-controls-and-ygrene-energy-fund-sign-nations-largestpace-energy-upgrade-project-with-metzler-real-estate.
- Larsen, P., C. A. Goldman, A. Satchwell. 2012a. "Evolution of the U.S. Energy Service Company Industry: Market Size and Project Performance from 1990-2008." *Energy Policy*, 50, 802-820.
- Larsen, P., C. A. Goldman, D. Gilligan, T. E. Singer. 2012b. "Incorporating Non-Energy Benefits into Energy Savings Performance Contracts." Proceedings of the ACEEE 2012 Summer Study on Energy Efficiency in Buildings. August. Accessed at:
- Larsen, P., E. Stuart, C. A. Goldman, D. Gilligan. 2014. Current Policies and Practices Related to the Incorporation of Non-energy Benefits in Energy Saving Performance Contract Projects. Proceedings of the ACEEE 2014 Summer Study on Energy Efficiency in Buildings. August. Accessed at: http://aceee.org/files/proceedings/2014/data/papers/8-320.pdf
- Linstone, H.A., and M. Turoff (Eds.), 1975. The Delphi method: Techniques and Applications. London: Addison-Wesley.
- NASEO (National Association of State Energy Offices). 2015. NASEO 2014 member survey.
- NCSL (National Conference of State Legislatures). 2013. State Energy Performance Contracting web page. November. Accessed at: http://www.ncsl.org/research/energy/state-energy-savings-performance-contracting.aspx.
- NORESCO. 2016. Performance Contracting [web page]. Accessed at: http://www.noresco.com/energy-services/en/us/solutions/Energy-Efficiency-Retrofits-Modernization/Performance-Contracting/.
- OMB (Office of Management and Budget). September 28, 2012. Memorandum for Heads of Executive Departments and Agencies: Addendum to OMB Memorandum M-98-13 on Federal Use of Energy Savings Performance Contracts (ESPCs) and Utility Energy Service

Contracts (UESCs) [Publication M-12-21]. Washington, D.C. Executive Office of the President. Accessed at: https://www.whitehouse.gov/sites/default/files/omb/memoranda/2012/m-12-21.pdf

- ORNL (Oak Ridge National Laboratory). 2016. Performance Contracting By State. Oak Ridge National Laboratory. Accessed at: http://web.ornl.gov/info/esco/legislation/newesco.shtml.
- Pace Nation. 2016. "Q1 2016 C-PACE Market Update." Accessed at: <u>http://www.pacenation.us/wp-content/uploads/2016/06/Market-update-Q1-2016.pdf</u>.
- Pew Charitable Trust. 2015. Fiscal 50: State Trends and Analysis. Accessed at: http://www.pewtrusts.org/en/multimedia/data-visualizations/2014/fiscal-50#ind4
- Rockwell, K. 2016. President's Performance Contracting Challenge: Performance Toward New \$4 Billion Goal. U.S. Department of Energy Federal Energy Management Program. Washington, D.C. May 15. Accessed at: http://energy.gov/eere/femp/downloads/presidentialperformance-contracting-challenge-performance-toward-new-4-billion
- Sacks, H. 2016, March 1. Personal communication with Howard Sacks, State of California Department of General Services.
- Satchwell, A., C. Goldman, P. Larsen, D. Gilligan, and T. Singer. 2010. "A Survey of the U.S. Energy Services Company (ESCO) Industry: Market Growth and Development from 2008 to 2011." Lawrence Berkeley National Laboratory 3479-E. June.
- Stuart E., Larsen, P., Goldman, C. A. & Gilligan, D. 2013. Current Size and Remaining Market Potential of the U.S. Energy Service Company Industry. Lawrence Berkeley National Laboratory 6300E. August.
- Stuart, E., Larsen, P., Goldman, C. A., & Gilligan, D. 2014. A method to estimate the size and remaining market potential of the U.S. ESCO (energy service company) industry. *Energy*, 77(December), 362-371. doi:10.1016/j.energy.2014.009.003.
- The Street. 2011. Ameresco reports fourth quarter and full year 2010 financial results. February 17. Accessed at: https://www.thestreet.com/story/11013767/1/ameresco-reports-fourth-quarter-and-full-year-2010-financial-results.html.
- The White House, Office of the Press Secretary. 2011. Presidential Memorandum Implementation of Energy Savings Projects and Performance-Based Contracting for Energy Savings. December 2. Accessed at http://www.whitehouse.gov/the-press-office/2011/12/02/presidential-memorandum-implementation-energy-savings-projects-and-perfo.
- The White House, Office of the Press Secretary. 2014. President Obama Announces Commitments and Executive Actions to Advance Solar Deployment and Energy Efficiency [fact sheet].

Accessed at: <u>https://www.whitehouse.gov/the-press-office/2014/05/09/fact-sheet-president-obama-announces-commitments-and-executive-actions-a</u>.

Vine, E. Nakagami, H., & Murakoshi, C. 1999. The evolution of the U.S. energy service company (ESCO) industry: from ESCO to super ESCO. *Energy 24, no. 6*(June), 479-492. Doi:10.1016/S0360-5442(99)00009-2.

Appendix A. Data Tables

Market	2008 (n=29) (\$ million)	2011 (n=35) (\$ million)	2014 (n=43) (\$ million)
Federal Govt.	\$ 583	\$1,102	\$1,073
State/Local Govt.	\$ 872	\$1,233	\$1,314
K–12 Schools	\$ 847	\$ 995	\$1,219
Univ./College	\$ 614	\$ 702	\$ 504
Healthcare	\$ 238	\$ 302	\$ 304
Housing/Other	\$ 356	\$ 385	\$ 342
Commercial/Industrial	\$ 277	\$ 419	\$ 409
TOTAL	\$3,786	\$5,138	\$5,165

Table A - 1. ESCO industry revenue (nominal \$) by market segment for 2008, 2011 and 2014

Table A - 2. ESCO industry revenue share by market segment for 2008, 2011 and 2014

Market	2008 (n=29)	2011 (n=35)	2014 (n=43)
Federal Govt.	15.4%	21.4%	20.7%
State/Local Govt.	23.0%	24.0%	25.4%
K–12 Schools	22.4%	19.4%	23.5%
Univ./College	16.2%	13.7%	10.0%
Healthcare	6.3%	5.9%	5.9%
Housing/Other	9.4%	7.5%	6.6%
Commercial/Industrial	7.3%	8.1%	7.9%
TOTAL	100.0%	100.0%	100.0%